\begin{table}%t3
%\centering
\par
\caption{\label{tab:par} Atmospheric parameters, as adopted from the
  literature, beryllium abundances as derived in this work, and lithium
  abundances, adopted from the literature when available or derived in
  this work when not. Details on the parameters and the abundances are
  given in the text. The reference papers of the atmospheric parameters
  and lithium abundances are also listed.} 
 \small \begin{tabular}{ccccccccccc} 
\hline\hline
Star &  $T_{\rm eff}$ & $\log~g$ &  $\log~g$ & $\xi$  & [Fe/H] &  Ref. & $\log$(Be/H) & $\log$(Be/H) & A(Li) & Ref. \\
   &     Kelvin     & Liter. & Paral. & km~s$^{-1}$ &  &  & 3131 &  average &  &   \\
\hline  
HIP 171 & 5275 & 4.10 & 4.30 & 1.05 & $-$0.90 & F00 & $-$11.55 & $-$11.51 & $\leq$0.80 & (1) \\
HIP 3026 & 5950 & 3.90 & 4.10 & 1.40 & $-$1.20 & F00 & $-$12.13 & $-$12.11 & 2.42 & (2) \\
HIP 7459 & 5909 & 4.46 & 4.59 & 1.23 & $-$1.15 & NS97 & $-$11.90 & $-$11.88 & 2.12 & (3) \\
HIP 10140 & 5425 & 4.10 & 4.25 & 0.85 & $-$1.00 & F00 & $-$11.68 & $-$11.64 & 1.54 & (2) \\
HIP 10449 & 5640 & 4.40 & 4.38 & 1.00 & $-$0.80 & F00 & $-$11.31 & $-$11.31 & 1.00 & (4) \\
HIP 11952 & 6100 & 4.20 & 4.25 & 0.95 & $-$1.60 & F00 & $-$12.28 & $-$12.26 & 2.16 & (2) \\
HIP 13366 & 5700 & 4.20 & 4.09 & 0.95 & $-$0.70 & F00 & $-$11.07 & $-$11.04 & 1.28 & (3) \\
HIP 14086 & 5075 & 3.60 & 3.48 & 1.10 & $-$0.60 & F00 & -- & -- & -- & -- \\
HIP 14594 & 5825 & 4.20 & 4.42 & 1.10 & $-$2.00 & F00 & $-$12.63 & $-$12.58 & 2.18 & (2) \\
HIP 17001 & 5360 & 3.00 & -- & 1.20 & $-$2.35 & GS88 & $\leq$$-$13.83 & -- & 1.17 & (2) \\
HIP 17147 & 5800 & 4.30 & 4.29 & 1.10 & $-$0.80 & F00 & $-$11.20 & $-$11.20 & 1.45 & (3) \\
HIP 18802 & 5886 & 4.33 & 4.22 & 1.38 & $-$0.85 & NS97 & $-$11.41 & $-$11.41 & 1.89 & (2) \\
HIP 19007 & 5150 & 4.50 & 4.74 & 1.20 & $-$0.50 & F00 & -- & -- & -- & -- \\ 
HIP 19814 & 5378 & 4.43 & -- & 0.84 & $-$0.69 & NS97 & -- & $\leq$$-$11.64 & 0.98 & (5) \\
HIP 21609 & 5200 & 3.80 & 4.55 & 1.55 & $-$1.60 & F00 & -- & -- & 1.43 & (2) \\
HIP 22632 & 5825 & 4.30 & 4.41 & 1.35 & $-$1.40 & F00 & $-$12.17 & $-$12.14 & 2.17 & (2) \\
HIP 24030 & 5700 & 4.20 & 4.26 & 1.11 & $-$1.20 & Pr00 & $-$11.55 & $-$11.53 & 2.08 & (3) \\
HIP 24316 & 5725 & 4.40 & 4.45 & 1.30 & $-$1.50 & F00 & $-$11.98 & $-$11.96 & 2.00 & (2) \\
HIP 31188 & 5750 & 4.10 & 4.28 & 1.65 & $-$0.70 & F00 & $-$11.46 & $-$11.46 & 1.95 & (4) \\
HIP 31639 & 5300 & 4.30 & 4.55 & 0.60 & $-$0.50 & F00 & $-$11.27 & $-$11.15 & $\leq$0.50 & (4) \\
HIP 33221 & 6097 & 4.09 & 4.00 & 1.86 & $-$1.30 & NS97 & $-$11.67 & $-$11.67 & 2.32 & (3) \\
HIP 33582 & 5725 & 4.30 & 4.32 & 1.25 & $-$0.50 & F00 & $-$11.09 & $-$11.09 & 1.10 & (4) \\
HIP 34285 & 5933 & 4.26 & 4.30 & 1.50 & $-$0.90 & NS97 & $-$11.85 & $-$11.85 & 2.22 & (3) \\
HIP 36491 & 5800 & 4.40 & 4.39 & 1.10 & $-$0.80 & F00 & $-$11.40 & $-$11.40 & 1.70 & (6) \\
HIP 36640 & 5830 & 3.64 & -- & 0.70 & $-$0.70 & GS91 & $-$11.44 & $-$11.42 & 2.36 & (3) \\
HIP 36818 & 5672 & 4.57 & 4.88 & 0.90 & $-$0.83 & NS97 & -- & $\leq$$-$12.58 & $\leq$0.70 & (3) \\
HIP 36849 & 5850 & 4.10 & 4.15 & 1.10 & $-$0.70 & F00 & $-$11.35 & $-$11.33 & 1.96 & (3) \\
HIP 37853 & 5822 & 4.42 & 4.18 & 1.21 & $-$0.78 & Ed93 & $-$11.16 & $-$11.19 & 1.37 & (3) \\
HIP 38625 & 5200 & 4.40 & 4.57 & 0.30 & $-$0.70 & F00 & $-$11.42 & -- & $\leq$0.16 & (7) \\
HIP 42592 & 6025 & 4.10 & 4.02 & 1.20 & $-$2.00 & F00 & $-$12.58 & $-$12.58 & 2.12 & (2) \\
HIP 44075 & 5900 & 4.20 & 4.11 & 1.25 & $-$0.70 & F00 & $-$11.12 & $-$11.12 & 2.06 & (3) \\
HIP 44124 & 5750 & 3.60 & -- & 0.10 & $-$1.80 & F00 & $-$12.92 & $-$12.86 & 1.96 & (4) \\
HIP 45554 & 6021 & 4.44 & 3.96 & 1.34 & $-$0.75 & NS97 & $-$11.45 & $-$11.45 & 2.09 & (3) \\
HIP 48152 & 6375 & 4.10 & 4.06 & 2.25 & $-$2.00 & F00 & $-$12.67 & $-$12.67 & 2.25 & (2) \\
HIP 50139 & 5600 & 4.30 & 4.30 & 0.35 & $-$0.70 & F00 & $-$11.28 & $-$11.23 & $\leq$0.40 & (4) \\
HIP 52771 & 5700 & 4.50 & 4.48 & 1.30 & $-$1.80 & F00 & -- & $-$12.55 & 2.16 & (2) \\
HIP 53070 & 5900 & 4.20 & 4.27 & 1.45 & $-$1.40 & F00 & $-$11.80 & $-$11.80 & 2.21 & (2) \\
HIP 55022 & 6450 & 4.20 & 4.05 & 1.80 & $-$0.80 & F00 & -- & $\leq$$-$12.75 & $\leq$1.31 & (2) \\
HIP 57265 & 5875 & 4.00 & 4.13 & 1.50 & $-$1.00 & F00 & $-$11.90 & $-$11.90 & 2.55 & (5) \\
HIP 58145 & 5946 & 4.41 & 3.96 & 1.32 & $-$1.04 & NS97 & $-$11.31 & $-$11.31 & 1.95 & (3) \\
HIP 58962 & 5831 & 4.36 & -- & 1.30 & $-$0.80 & NS97 & $-$11.68 & $-$11.64 & 1.95 & (3) \\
HIP 59490 & 6046 & 4.46 & 4.46 & 1.34 & $-$1.26 & NS97 & $-$10.60 & $-$10.58 & 2.51 & (3) \\
HIP 59750 & 6200 & 4.40 & 4.26 & 1.10 & $-$0.60 & F00 & -- & $\leq$$-$12.50 & $\leq$1.10 & (3) \\
HIP 60632 & 6200 & 4.40 & 4.45 & 1.35 & $-$1.50 & F00 & $-$12.10 & $-$12.10 & 2.21 & (2) \\
HIP 62882 & 5600 & 3.70 & -- & 0.04 & $-$1.10 & F00 & $-$11.20 & $-$11.20 & 2.19 & (8) \\
HIP 63559 & 5865 & 4.41 & 4.16 & 1.26 & $-$0.93 & NS97 & -- & -- & 2.14 & (2) \\
HIP 63918 & 5720 & 4.14 & 3.91 & 1.49 & $-$0.65 & NS97 & $-$10.95 & $-$10.95 & 1.99 & (3) \\
HIP 64426 & 5800 & 4.10 & 4.10 & 1.25 & $-$0.70 & F00 & $-$11.31 & $-$11.31 & 2.01 & (3) \\
HIP 66665 & 5500 & 3.80 & 3.77 & 1.05 & $-$0.80 & F00 & $-$11.61 & $-$11.63 & 0.90 & (4) \\
HIP 67655 & 5396 & 4.38 & 4.62 & 0.92 & $-$0.93 & NS97 & $-$11.69 & $-$11.50 & $\leq$1.25 & (2) \\
HIP 67863 & 5686 & 4.40 & 4.42 & 1.13 & $-$0.70 & NS97 & $-$10.95 & $-$10.95 & $\leq$1.34 & (2) \\
HIP 70681 & 5450 & 4.50 & 4.55 & 0.80 & $-$1.10 & F00 & $-$11.12 & $-$11.12 & 1.54 & (3) \\
HIP 71458 & 5480 & 3.10 & -- & 1.98 & $-$2.10 & McW95 & $\leq$$-$13.90 & -- & 1.29 & (2) \\
HIP 72461 & 5875 & 4.10 & 4.31 & 0.40 & $-$2.30 & F00 & -- & -- & 2.22 & (2) \\
HIP 74067 & 5575 & 4.30 & 4.41 & 1.10 & $-$0.80 & F00 & $-$11.36 & $-$11.36 & 0.95 & (4) \\
HIP 74079 & 5825 & 4.00 & 3.92 & 1.30 & $-$0.70 & F00 & $-$11.28 & $-$11.28 & 2.24 & (2) \\
HIP 77946 & 6550 & 3.60 & 3.70 & 1.65 & $-$0.90 & F00 & $\leq$$-$12.95 & $\leq$$-$12.85 & $\leq$1.45 & (6) \\
HIP 80003 & 5486 & 4.80 & 4.80 & 0.00 & $-$0.87 & SB02 & $-$11.62 & $-$11.62 & 1.25 & (5) \\
HIP 80837 & 5800 & 4.10 & 4.07 & 1.15 & $-$0.70 & F00 & $-$11.22 & $-$11.20 & 1.53 & (9) \\
\hline
HIP 81170 & 5175 & 4.70 & 4.64 & 0.30 & $-$1.10 & F00 & $-$11.53 & $-$11.42 & $\leq$0.30 & (10) \\
HIP 85963 & 6227 & 3.94 & 3.75 & 2.16 & $-$0.71 & Ed93 & $-$11.78  & $-$11.78  & $\leq$0.70  & (4) \\
HIP 87693 & 6175 & 4.00 & 3.99 & 0.90 & $-$2.00 & F00 & $-$12.77 & $-$12.77 & 2.20 & (2) \\
HIP 88010 & 5200 & 4.00 & 4.15 & 0.70 & $-$1.40 & F00 & $-$12.40 & $-$12.34 & 1.67 & (2) \\
HIP 92781 & 5650 & 4.20 & 4.12 & 0.95 & $-$0.60 & F00 & $-$11.14 & $-$11.14 & -- & -- \\
HIP 94449 & 5625 & 3.70 & 3.74 & 1.15 & $-$1.20 & F00 & $-$11.50 & $-$11.50 & 1.81 & (6) \\
HIP 98020 & 5325 & 4.60 & 4.54 & 1.10 & $-$1.60 & F00 & -- & -- & 1.61 & (2) \\
HIP 98532 & 5550 & 3.60 & -- & 1.30 & $-$1.10 & F00 & $-$11.43 & $-$11.43 & 2.15 & (3) \\
HIP 100568 & 5650 & 4.40 & 4.51 & 1.10 & $-$1.00 & F00 & $-$11.98 & $-$11.92 & 1.92 & (2) \\
HIP 100792 & 5875 & 4.20 & 4.26 & 1.40 & $-$1.10 & F00 & $-$11.97 & $-$11.93 & 2.02 & (2) \\
HIP 101346 & 6000 & 3.90 & 3.74 & 1.40 & $-$0.50 & F00 & $-$11.52 & $-$11.52 & 2.15 & (9) \\
HIP 103498 & 5894 & 4.38 & 4.29 & 1.32 & $-$1.03 & Ed93 & $-$11.36 & $-$11.39 & 2.00 & (11) \\
HIP 104660 & 5500 & 3.90 & 4.01 & 1.15 & $-$0.80 & F00 & $-$11.41 & $-$11.44 & 1.10 & (2) \\
HIP 105858 & 6139 & 4.34 & 4.26 & 1.57 & $-$0.67 & Ed93 & $-$11.32 & $-$11.32 & 2.39 & (3) \\
HIP 105888 & 5700 & 4.00 & 3.99 & 1.00 & $-$0.60 & F00 & $-$11.13 & $-$11.13 & 1.37 & (6) \\
HIP 106447 & 6089 & 4.04 & -- & 1.50 & $-$2.48 & SB02 & $-$12.88 & -- & 2.23 & (5) \\
HIP 107975 & 6275 & 3.90 & 3.87 & 1.50 & $-$0.50 & F00 & $-$12.25 & $-$12.25 & $\leq$1.10 & (9) \\ 
HIP 108490 & 6009 & 4.41 & 4.30 & 1.37 & $-$0.72 & Ed93 & $-$11.34 & $-$11.30 & 2.31 & (3) \\
HIP 109067 & 5300 & 4.30 & 4.68 & 0.85 & $-$0.80 & F00 & $-$11.39 & $-$11.37 & $\leq$0.10 & (4) \\
HIP 109558 & 6025 & 4.00 & 4.09 & 1.10 & $-$1.50 & F00 & $-$12.30 & $-$12.34 & 2.10 & (2) \\
HIP 109646 & 5910 & 4.25 & 4.21 & 1.50 & $-$0.64 & Ed93 & $-$11.38 & $-$11.36 & 2.23 & (3) \\
HIP 112229 & 5983 & 4.37 & 4.16 & 1.40 & $-$0.65 & Ed93 & $-$11.21 & $-$11.25 & 2.31 & (3) \\
HIP 114271 & 6200 & 4.10 & 4.21 & 0.90 & $-$1.70 & F00 & $-$12.44 & $-$12.39 & 2.32 & (2) \\
HIP 114962 & 5825 & 4.30 & 3.81 & 1.40 & $-$1.40 & F00 & $-$12.40 & $-$12.34 & 2.25 & (2) \\
HIP 115167 & 6100 & 3.80 & 3.57 & 1.15 & $-$1.50 & F00 & $-$12.45 & $-$12.45 & 2.17 & (2) \\
HIP 117041 & 5300 & 4.20 & 4.11 & 0.90 & $-$0.80 & F00 & $-$11.34 & $-$11.34 & $\leq$0.10 & (4) \\
G05-19 & 5942 & 4.24 & -- & 0.89 & $-$1.10 & SB02 & $-$11.98 & $-$12.01 & 2.26 & (5) \\
G05-40 & 5863 & 4.24 & -- & 1.48 & $-$0.83 & NS97 & $-$11.15 & $-$11.15 & 1.90 & (3) \\
G66-51 & 5255 & 4.48 & -- & 0.90 & $-$1.00 & Pr00 & $-$11.47 & $-$11.37 & $\leq$0.30 & (4) \\
G166-37 & 5300 & 4.80 & 4.69 & 0.60 & $-$1.20 & F00 & -- & -- & 1.28 & (5) \\
G170-21 & 5664 & 4.65 & -- & 0.00 & $-$1.45 & SB02 & $-$11.65 & $-$11.58 & 1.91 & (5) \\
\hline
\end{tabular}
\medskip
(1) Takeda \& Kawanomoto (\cite{TK05}); (2) Charbonnel \& Primas 
(\cite{CP05}); (3)~Chen et~al. (\cite{Ch01}); (4)~this work; (5)~Boesgaard et~al. (\cite{Bo05}); (6)~Fulbright (\cite{F00}); (7)~Favata et~al. (\cite{Fav96}); (8)~Ryan \& Deliyannis (\cite{RyD95});
 (9)~Romano et~al. (\cite{Ro99}); (10)~Spite et~al. (\cite{Sp94}); (11)~Gratton et~al. (\cite{Gr00}). 
\end{table}